Author's response to reviews

Title: Treatment of two postoperative endophthalmitis cases due to Aspergillus flavus and Scopulariopsis spp. with local and systemic antifungal therapy

Authors:

Sayime Aydin (sayimeaydin@mynet.com)
Bulent Ertugrul (mbulentertugrul@yahoo.com)
Berna Gultekin (berngul@yahoo.com)
Guliz Uyar (gulizuyar@yahoo.com)
Erkin Kir (erkinkir@yahoo.com)

Version: 4 Date: 14 February 2007

Author's response to reviews:

Response to reviewer;

Major compulsory revisions:

1. The title, abstract and conclusion sections were modified based on the comments of the reviewer.
2. The abstract was shortened.
3. "Parasites" were excluded from abstract and background.
4. "light of the literature" were excluded from background.
5. The total number of grams were removed.
6. The second case had excessive cachexia, hypoproteinemia and hypoalbuminemia. We administered low dose of voriconazole as to prevent side-affects. However hepatic enzymes elevated on the tenth day of the treatment and we had to stop systemic voriconazole.

Minor essential revision:

1. The sentence was changed as follows: 
   ".... it affects the vision seriously."